**Author's response to reviews**

**Title:** What factors explain the number of physical therapy treatment sessions in patients referred with low back pain; a multilevel analysis

**Authors:**

Ilse CS Swinkels (i.swinkels@nivel.nl)
Raymond H Wimmers (r.wimmers@hartstichting.nl)
Peter P Groenewegen (p.groenewegen@nivel.nl)
Wil JH van den Bosch (w.vandenbosch@hag.umcn.nl)
Joost Dekker (j.dekker@vumc.nl)
Cornelia HM van den Ende (e.vandenende@cukz.umcn.nl)

**Version:** 5 **Date:** 21 November 2005

**Author's response to reviews:**

Dear editor,

According to your instructions, I have transposed the website addresses from the manuscript text to the reference list. Furthermore, I have listed all authors' names in the reference list instead of using the term "et all".

With kind regards,

Ilse Swinkels